

# G OPEN ACCESS

**Citation:** Gimmelli R, Papoff G, Saccoccia F, Lalli C, Gemma S, Campiani G, et al. (2024) Effects of structurally distinct human HDAC6 and HDAC6/ HDAC8 inhibitors against *S. mansoni* larval and adult worm stages. PLoS Negl Trop Dis 18(2): e0011992. https://doi.org/10.1371/journal. pntd.0011992

Editor: Aaron R. Jex, Walter and Eliza Hall Institute of Medical Research, AUSTRALIA

Received: August 23, 2023

Accepted: February 13, 2024

Published: February 28, 2024

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0011992

**Copyright:** © 2024 Gimmelli et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the manuscript and the <u>supporting</u> information.

RESEARCH ARTICLE

# Effects of structurally distinct human HDAC6 and HDAC6/HDAC8 inhibitors against *S. mansoni* larval and adult worm stages

Roberto Gimmelli<sup>1,2</sup>, Giuliana Papoff<sup>1</sup>, Fulvio Saccoccia<sup>1</sup>, Cristiana Lalli<sup>1</sup>, Sandra Gemma<sup>3</sup>, Giuseppe Campiani<sup>3</sup>, Giovina Ruberti<sup>1</sup>\*

1 Institute of Biochemistry and Cell Biology, National Research Council (IBBC-CNR), Adriano Buzzati-Traverso Campus, Monterotondo, Rome, Italy, 2 Dipartimento di Scienze Biochimiche "A. Rossi Fanelli", Sapienza Università di Roma, Roma, Italy, 3 Department of Biotechnology, Chemistry and Pharmacy, University of Siena, Siena, Italy

\* giovina.ruberti@cnr.it

## Abstract

Schistosomiasis is a major neglected parasitic disease that affects more than 240 million people worldwide caused by Platyhelminthes of the genus Schistosoma. The treatment of schistosomiasis relies on the long-term application of a single safe drug, praziguantel (PZQ). Unfortunately, PZQ is very effective on adult parasites and poorly on larval stage and immature juvenile worms; this can partially explain the re-infection in endemic areas where patients are likely to host parasites at different developmental stages concurrently. Moreover, the risk of development of drug resistance because of the widespread use of a single drug in a large population is nowadays a serious threat. Hence, research aimed at identifying novel drugs to be used alone or in combination with PZQ is needed. Schistosomes display morphologically distinct stages during their life cycle and epigenetic mechanisms are known to play important roles in parasite growth, survival, and development. Histone deacetylase (HDAC) enzymes, particularly HDAC8, are considered valuable for therapeutic intervention for the treatment of schistosomiasis. Herein, we report the phenotypic screening on both larvae and adult Schistosoma mansoni stages of structurally different HDAC inhibitors selected from the in-house Siena library. All molecules have previously shown inhibition profiles on human HDAC6 and/or HDAC8 enzymes. Among them we identified a quinolone-based HDAC inhibitor, NF2839, that impacts larval and adult parasites as well as egg viability and maturation in vitro. Importantly, this guinolone-based compound also increases histone and tubulin acetylation in S. mansoni parasites, thus representing a leading candidate for the development of new generation anti-Schistosoma chemotherapeutics.

## Author summary

Neglected tropical diseases, such as schistosomiasis, which is caused by parasitic worms, represent serious public health problems in many countries around the world.

Funding: This work was supported by the Ministero dell'Università e della Ricerca (MUR), PRIN Project No. 20154JRJPP to G.C. and G.R. and by the CNR (National Research Council)-CNCCS (Collezione Nazionale di Composti Chimici e Centro di Screening) "Rare, Neglected and Poverty Related Diseases Schistodiscovery Project" (DSB.AD011.001.003 to G. R.) and by the Lazio Innova POR FESR Lazio 2014 to 2020 HDACiPLAT Project (grant number A0375-2020-36575 to G. R.) The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RG received salary for postdoctoral fellowship from Lazio Innova grant.

**Competing interests:** The authors have declared that no competing interests exist.

*Schistosoma mansoni* is the major parasitic platyhelminth species causing intestinal schistosomiasis. Currently the control of schistosomiasis is heavily reliant on the drug praziquantel (PZQ) that it is unfortunately not highly effective against larval and immature worm stages. The use of PZQ in mass treatment programs means that the development of resistance is likely, therefore identifying novel drug targets and their associated compounds is critical. Drugs that inhibit enzymes that modify the chromatin structure have been developed for several diseases. We and others have shown that *S. mansoni* epigenetic enzymes are also potential therapeutic targets. Here, we evaluate and characterize several histone deacetylases (HDAC) inhibitors on the larval and adult stage of *Schistosoma mansoni* and report the anti-parasitic profile of these compounds. Some of the HDAC inhibitors tested showed potent effects, particularly NF2839, that impacts larval and adult parasites as well as egg viability and maturation. The findings of this study provide a starting point for the development of new HDAC-based schistosomicidal compounds.

#### Introduction

Parasitic diseases remain a major cause of morbidity and mortality globally, mostly affecting people living in the poorest regions of the world. Changes in the environment and population dynamics pose new and global challenges also for schistosomiasis, a neglected tropical disease caused by Platyhelminthes of the genus Schistosoma. The three most relevant species for human infections are Schistosoma mansoni, Schistosoma hematobium, and Schistosoma japoni*cum* [1]. Currently, there is no vaccine available to prevent human schistosomiasis, its development being proven to be extremely challenging, mainly for the complexity of the schistosome life cycle and for the parasite's ability to evade the host immune system [2]. Praziguantel (PZQ) is the only approved drug to treat schistosomiasis. PZQ is very effective against adult worms; it is safe and well tolerated, with no significant adverse effects; unfortunately, it is poorly effective on immature stages, such as larval and juvenile worms [3]. PZQ is largely used in mass drug administration programs for the control of schistosomiasis in all endemic settings and development of drug-resistance strains represents a constant threat, also for the appearance of parasite hybrids. Climate changes can further enhance the potential for human and animal populations to encounter new infectious agents, and co-infection by multiple pathogen species within the same host could increase. In the case of helminth parasites, this can occur also through heterospecific (between-species) mate pairings, which can lead to the formation of hybrid offspring. Hybridizations, as well as subsequent introgressions increase genetic diversity and can have important implications in terms of disease control and outcome [4]. Schistosomes are known to hybridize, and natural hybrids are more and more frequently found [5-7]. For all above reasons, new drugs able to target multiple stages of the parasite's life cycle and multiple species are urgently needed. The use of epigenetic drugs including histone deacetylases (HDAC) inhibitors were suggested as a potential novel interesting strategy for antiparasitic therapy including schistosomiasis [8].

HDACs are members of a family of proteins highly conserved across all eukaryotes. Their main action is to detach acetyl groups from lysine residues at the N-terminal tail of DNA-binding of histones and non-histone proteins [9]. Both Class I (HDACs 1, 3, and 8) and Class II orthologues (HDACs 4, 5, 6/10, 9) have been identified in *S. mansoni* [10–12]. SmHDAC8 is a class I zinc dependent HDAC, which was demonstrated to be abundantly expressed in all *S. mansoni* life cycle stages [10]. Noteworthy, the human homologue hHDAC8 is poorly expressed in human healthy cells and SmHDAC8 is less conserved compared to its human orthologue than other class I schistosome HDACs 1 and 3 [13]. Therefore, SmHDAC8 has been proposed as a promising target for drug discovery [10]. In recent years, both our group and other research teams have identified HDAC inhibitors that demonstrated effectiveness against *S. mansoni* and/or impacted the enzymatic activity of *S. mansoni* HDAC8 (*Sm*HDAC8) [13–22]. Most discovered inhibitors active on SmHDAC8 present a zinc-binding group, such as a hydroxamate or a sulfur moiety [17]. The general pharmacophoric model of HDAC inhibitors have three key components namely: the surface recognition group (cap), the zinc binding group (ZBG) and a linker bridging the two portions. To achieve better SmHDAC8 activity and selectivity, different structural modifications of the cap and linker have been investigated. Examples of hydroxamic acid-based SmHDAC8 inhibitors reported in literature are shown in Fig 1 [16–18,21–22]

Here to search for novel valuable molecules, we tested structurally different HDAC inhibitors, chosen from the Siena in-house library [23–26]. Fourteen HDAC inhibitors, already recognized to be active on mammalian cells and with a known profile of inhibitory activity on recombinant human HDAC1, HDAC6 and HDAC8, were selected to be tested in phenotypic screenings against *S. mansoni* parasites and SmHDAC8 recombinant protein activity. Selection of novel hits will facilitate further development of novel schistosomicidal molecules.

#### Methods

#### **Ethics statement**

Animal work was approved by the National Research Council, Institute of Biochemistry and Cell Biology Animal Welfare Committee (OPBA) and by the competent authorities of the Italian Ministry of Health, DGSAF, Rome (authorization nos. 25/2014-PR and 336/2018-PR). All experiments were conducted in respect to the 3R rules according to the ethical and safety rules and guidelines for the use of animals in biomedical research provided by the relevant Italian law and European Union Directive (Italian Legislative Decree 26/2014 and 2010/63/EU) and the International Guiding Principles for Biomedical Research Involving Animals (Council for the International Organizations of Medical Sciences, Geneva, Switzerland).

#### Chemistry

HDAC inhibitors used in this study were prepared as previously reported [23-26].

#### Reagents

Chemicals and molecular biology reagents if not otherwise stated were purchased from Merck Life Science Srl (Milan, Italy) or Thermo Fisher Scientific, Italy; Talon superflow histidine-tagged protein purification resin was from GE Healthcare Life Sciences; CellTiter-Glo reagent from Promega; the primary monoclonal anti-α-tubulin antibody (DM1A) and anti-acetylated tubulin (6-11B-1) from Sigma–Aldrich; the anti-acetylated-lysine (Ac-K2-100) from Cell Signaling Technology; the goat anti-mouse and anti-rabbit IgG (H+L) horseradish peroxidase secondary antibodies from Bio-Rad Laboratories. Fluor De Lys HDAC8 Fluorimetric Drug Discovery Kit (BML-AK518) was from Enzo Life Science. Tissue culture media reagents were from Gibco (ThermoFisher Scientific) or EuroClone SpA, Italy.

#### Mammalian cell lines and viability assays

BJ cells human fibroblast established from normal foreskin of a neonatal male from the American Type Culture Collection (ATCC; Manassas, VA, USA) (ATCC\_CRL-2522) were culture in Minimal Essential Medium (ThermoFisher Scientific, Gibco-11090-081) supplemented with



**Fig 1. Examples of hydroxamic acid based SmHDAC8 inhibitors reported in literature.** J1038, TH65 [16,17]; Compound 9 [22]; NF2624, NF2886 [18,21] are shown.

https://doi.org/10.1371/journal.pntd.0011992.g001

10% heat-inactivated fetal bovine serum (FBS, EuroClone SpA, Milan, Italy), L-glutammine 2 mM, Sodium Pyruvate 1 mM (GIBCO 11360–039), 0,1 mM non-essential amino acids (NEAA 100x GIBCO- 11140–035) and 100 U/ml of penicillin/streptomycin. The NCTC clone 929 (L-929) murine fibroblast isolated from subcutaneous connective tissue (ATCC-CCL-1) were cultured in Dulbecco's Modified Eagle Medium (DMEM) supplemented with 10 mM Hepes (pH 6.98–7.30), 1 mM l-glutamine, 100 U/ml of penicillin/streptomycin (BioWhittaker, Lonza, EuroClone SpA), and heat-inactivated 10% FBS. All cells were cultured at 37°C in a 5% CO2 humidified incubator.

The MTT method [27] was employed as previously described to determine the viability of fibroblast cell lines [28]. Briefly, cells seeded into 96-well plates at a density of 20,000 cells/well after overnight incubation, were exposed to selected compounds at the indicated concentrations for 24–72 h. Next they were gently washed, incubated with MTT (bromuro di 3-(4,5-dimetiltiazol-2-il)-2,5-diphenyltetrazolium) for 4 h, and processed for color detection with DMSO. The resulting purple solution was spectrophotometrically measured at 570 nm using a Varioskan Lux instrument and Skanit software (Thermo Fisher Scientific).

The optical density values for both assays were expressed as a percentage of cell survival and normalized with the value of cells treated with vehicle (DMSO), and the data were analyzed using GraphPad Prism v9.5.1 software (San Diego, CA, USA).

#### Life cycle S. mansoni maintenance, viability assays and egg counts

A Puerto Rican strain of *S. mansoni* was maintained by cycling within albino *Biomphalaria glabrata*, as the intermediate host, and ICR (CD-1) outbred female mice as the definitive host, as previously described [29]. BioWhittaker DMEM lacking phenol red and containing 4500 mg/l glucose, 1 mM Hepes pH 6.98–7.30, 2 mM L-glutamine, 1x antibiotic-antimycotic reagent (Life Technologies) and 10% heat inactivated FBS was used as tissue culture medium for *Schistosoma mansoni* newly transformed schistosoma (NTS). The NTS obtained by

mechanical transformation of cercariae and Percoll gradient was used for the ATP-based luminescence viability assays with the CellTiter-GLO as previously described [29–30]. Briefly, the NTS (150/well) were incubated in DMEM medium (w/o phenol red) complete medium with compounds dissolved in DMSO for 72 h, 37°C in 5% CO<sub>2</sub>, in 96-well black plates. Then, 50 µl/well of CellTiter-GLO reagent was added and the relative luminescence unit signal was recorded after 30 min with Varioskan Lux. Data analysis was achieved using GraphPad Prism v9.5.1 software (GraphPad Software Inc) and LD<sub>50</sub> was obtained after nonlinear regression curve fitting, according to log (inhibitor) *versus* normalized response with a variable slope curve model.

For studies on adult worm pairs, mice infected 7-8 weeks previously with double sex cercariae were euthanized with intraperitoneal injections of Tiletamine/Zolazepam (800 mg/kg) + Xylazine (100 mg/kg) and adult parasites were harvested by reversed perfusion of the hepatic portal system and mesenteric veins. Adult worms were cultured in DMEM containing phenol red and with 4500 mg/l glucose, 2 mM L-glutamine, 100 U/ml penicillin, 100 µg/ml streptomycin, 1x antibiotic-antimycotic reagent, and 10% heat inactivated FBS. Viability assays on adult worm pairs were based on a phenotyping scoring assessment as previously described [29,31]. Briefly, five adult pairs were incubated with selected compounds in 3 ml of DMEM complete medium. For each compound, three experiments were performed, and compounds were given to parasites only once without medium addition and/or replacement. DMSO (vehicle)-treated worms were used as control samples. Viability was monitored under a Leica Model MZ12 stereomicroscope for 7 days and viability scores (0-3) were assigned. In particular, the following scoring criteria were adopted: 3, denoting plate-attached, good movements, clear aspect; 2, representing slower or diminished movements, darkening, minor tegumental damage; 1, denoting movements heavily lowered, darkening, tegument heavily damaged; and 0, meaning dead, lack of any movement. For each sample, the total score was determined by the ratio of the sum of worm scores for the total number of worms.

The eggs produced by worm pairs *in vitro* were counted at day 3 upon treatment using an inverted Leica DM IL microscope (Leica Microsystems, Wetzlar, Germany). Images were captured with a BX41 Olympus (Olympus, Tokyo, Japan) microscope and a brightfield objective 10× served by an Olympus DP23 microscope digital camera and visualized using "CellSens Entry" software (version 3.1.1). Egg maturation morphological score was assigned based on the Vogel and Prata's staging system of egg maturation previously reported [32–33].

#### Confocal microscopy analysis

Carmine Red staining was performed as previously described [30]. Images were acquired on a confocal laser scanning TCS SP5 microscope (Leica Microsystems, Wetzlar, Germany), using a 40x (NA = 1.25) oil-immersion lens with optical pinhole at 1AU; Argon laser at 488 nm was used as excitation source and fluorescence was recovered in the range of 500–700 nm. Images were collected as a single stack.

#### Western blot analysis

Ten adult pairs were treated with selected HDAC inhibitors for 48 h or 72 h at partially lethal and sublethal concentrations. Compound concentrations were selected based on the viability curve, Trichostatin A (TSA), a pan-HDAC inhibitor, and NF2503 a SmHDAC8 inhibitor chemically characterized by the presence of a hydroxamate-based MBG (metal binding group) coupled to a tricyclic thieno[3,2-b]indole core as capping group, previously described [18] were used as positive controls in the experiments; DMSO (vehicle) was the negative control. Parasites were processed as previously described [18] and samples were analyzed by 15% SDS-PAGE, and immunoblotting was performed using  $\alpha$ -tubulin DM1A (1:5000),  $\alpha$ -

acetylated-lysine (1:4000) and  $\alpha$ -acetylated tubulin (1:3000) antibodies. A Chemidoc XRS Bio-Rad with a chemiluminescent camera and Bio-Rad ImageLab 4.0 software were used for the acquisition and analysis of images. Each Western blot analysis was calibrated by Ponceau staining.

#### E. coli protein purification and SmHDAC8 activity and inhibition assays

SmHDAC8 protein was produced and purified by a method previously described by us [18]. Inhibition assays of SmHDAC8 were performed as already described in previous publications [18,21]. Briefly, the commercial Fluor de Lys kit (BML-KI178) of Enzo Life Science was used for testing the residual activity of the recombinant enzyme upon preincubation with each of the selected compounds. First the enzyme was preincubated with the selected compounds 15 min before substrate addition to the mixture. The Fluor de Lys substrate was used at 50  $\mu$ M. Then, compounds, substrate, and enzyme were incubated together, and the reaction was allowed to proceed for 1 h at 30°C; subsequently, 2  $\mu$ M TSA within 50  $\mu$ L of 1× Developer II was added, and the mixture was further incubated for 1 h at 30°C to quench the reaction. Fluorescence was measured in a plate reader (Varioskan Lux, Thermo Fisher Scientific) with excitation at  $\lambda = 370$  nm and emission at  $\lambda = 450$  nm.

#### Statistical analysis

All statistical tests were performed using GraphPad Prism version 9.5.1 software (San Diego, CA, USA). The data are shown as mean  $\pm$  standard deviation (SD) or Standard error of the mean (SEM) as indicated.

#### **Results and Discussion**

Numerous HDAC inhibitors with a clear inhibition profile on human HDAC enzymes showed no effect on the parasite, which may be due to their selectivity for human targets or more likely to poor bioavailability or stability in the parasites; therefore, phenotypic assays on *S. mansoni* are important to identify valuable compounds. Here, we evaluated the effects on NTS (*S. mansoni* larval stage) of a set of inhibitors proven to be active on human class I (hHDAC1 and hHDAC8) and/or class II (hHDAC6) enzymes (Fig 2) and belonging to the in-house Siena library of HDAC inhibitors [23–26].

Compounds 1–3 were chosen among the most selective hHDAC6 inhibitors with an indoline-based cap group and different aromatic and aliphatic moieties at the C2 and C3 positions respectively. These compounds were shown having low inhibitory potency over hHDAC1 and hHDAC8 [23]. Compounds 4–6 were selected among the spiroindoline-based HDAC6 isoform-selective inhibitors [24]. Compounds 7–9 are potent and selective quinolone-based hHDAC6 inhibitors [25]. Compounds 10–14 are dual hHDAC6/hHDAC8 inhibitors with a diphenyl-azetidin-2-one scaffold [26].

#### Effects of HDAC inhibitors 1-14 on S. mansoni larval stage

The HDAC inhibitors were screened against NTS treated at 50  $\mu$ M concentration for 72 h to determine their ability to impair viability by using the ATP-based luminescence assay previously developed and validated in our laboratory [29,30]. Two out of 14 compounds, both presenting a quinolone-based cap group, were shown to be effective *in vitro*, decreasing the NTS viability by at least 50%. Next, to calculate the LD<sub>50</sub> value, dose–response curves were designed by treating parasites with compounds 7 (NF2836) and 9 (NF2839) at concentrations ranging from 0.7  $\mu$ M up to 100  $\mu$ M. Compound 8 (NF2838) with a similar structure was assayed as





NF2376(1)

Spiroindoline scaffold (ref. 24)



NF2657(4)

Quinolone scaffold (ref. 25)



NF2836 (7)

но-<sup>Н</sup> NF2838 (8)

HO

HO

NF2451 (2)

NF2866 (5)



NF2870 (6)

NF2454 (3)



NF2839 (9)

Azetidin-2-one scaffold (ref. 26)



NF2817 (10)



NF2853 (13)



HONN

NF2819 (12)



NF2921 (14)

Fig 2. Structure, type of cap, and original reference of HDAC inhibitors tested against S. mansoni.

https://doi.org/10.1371/journal.pntd.0011992.g002



Fig 3. Schistosome viability assays. Panel A: Representative dose-response curves of the inhibitors on NTS. The y-axis indicates the percentage of viability normalized against DMSO (100%) and gambogic acid 10  $\mu$ M (0%). The calculated LD<sub>50</sub> ( $\mu$ M) average +/- SD are shown. Panel B: dose-response curve of the inhibitors on adult schistosome worm pairs. DMSO (vehicle) was used as negative control (100% viability), the indicated compounds were assayed at 5 (full triangle) 10 (inverted full triangle), 20 (full square) and 50  $\mu$ M (full circle). Data are expressed as a % severity score (viability) relative to DMSO. Each point represents the average ± standard deviation of three independent experiments.

https://doi.org/10.1371/journal.pntd.0011992.g003

well at the same concentrations. The  $LD_{50}$  values of the three compounds are reported in Fig 3A, along with dose–response curves. The most active compound is NF2839 with a  $LD_{50}$  of 10.6 +/- 3.9  $\mu$ M while the NF2836 shows a lower potency, and the NF2838, with a similar structure, is completely inactive.

# Effects of the quinolone-based compounds on the viability of adult worm pairs, egg production, viability, and maturation

The initial screening of compounds on NTS aimed at identifying promising hit compounds able to impair the larvae viability with a good potency. Since different stages of the schistosome life cycle have different susceptibility to drug actions, the activity of NF2836, NF2838 and NF2839 was investigated also against adult *S. mansoni* worm pairs. Parasites were treated initially at 20  $\mu$ M and 50  $\mu$ M; their impact on viability was monitored for 7 days upon treatment. For the most active one, NF2839, parasite's viability was also assessed at lower concentrations 5 and 10  $\mu$ M. Viability of each schistosome was scored from 3 (no effect) to 0 (severe effects) by assigning daily worm phenotype scores (plate attachment, movement, color, gut peristalsis,

tegument damage) as previously described [29,31] (Fig 3B). Only one of the three compounds markedly decreases the viability of treated worms. NF2839 killed all parasites at 50  $\mu$ M in 72 h and caused 70% death of *S. mansoni* adult worm pairs in 72 h also at 20  $\mu$ M. The other two quinolone compounds caused only a 50% reduction in viability at 50  $\mu$ M; we cannot exclude that their lower efficacy is due to poor bioavailability or stability in our experimental setting or to targeting of HDAC not impacting viability.

The most active compound, NF2839, and one of the less active quinolone compounds, NF2836, in killing adult worm pairs, were then evaluated for their impact on egg production and maturation *in vitro* (Fig 4). First, the highest sublethal concentrations that ensured 50–100% viability and pairing were identified for each compound; hence, egg counts were performed at 72 h upon treatment and normalized to the number of couples. Worm pairs were treated with NF2839 at 5, and 10  $\mu$ M concentrations; or NF2836 at 20 and 50  $\mu$ M concentrations. While we did not observe a decrease in the number of eggs laid by worm pairs treated



**Fig 4. Quinolone-based HDAC inhibitors impair egg viability and maturation.** Panel A. Representative microscopy images of *S. mansoni* eggs laid by worm pairs exposed to DMSO, NF2836 (20 or 50  $\mu$ M) or NF2839 (5, 10  $\mu$ M). Panel B. Histograms represent the percentages of eggs at different maturation stages +/- SEM counted at 72 h after treatment; 5–6 microscopy images of at least 3 independent experiments were analyzed. Filled red triangles indicate viable eggs at stages I–II (immature); empty red triangles indicate damaged/morphological abnormal eggs at stages III-V (intermediate/developed); empty red arrows indicate damaged/morphological abnormal eggs at stages III-V; ef, egg fragments; sp, sperm cells; oc, oocyte. Scale Bars = 200  $\mu$ m.

https://doi.org/10.1371/journal.pntd.0011992.g004

with the inhibitors, the eggs showed clear signs of damage and defects in maturation when compared with the vehicle control samples. Moreover vitelline, germ cells and egg fragments were present in the culture medium. Therefore at 72 h upon worm pairs treatment the eggs were counted and classified by microscopic observation according to the Vogel and Prata staging system of egg maturation [32,33] and grouped as immature (stages I and II), intermediate/ developed (stages III–V), viable or damaged eggs. The samples treated with vehicle (DMSO) developed as expected. A drastic effect of NF2836 and NF2839 compounds on egg maturation was observed; essentially all eggs were arrested in stage I/II. Moreover, following treatment with the compounds, eggs appeared seriously damaged and not viable (Fig 4).

To characterize the phenotypic alterations induced by compound treatment in more detail, carmine-red staining, followed by confocal laser scanning microscopy (CLSM) analysis was performed (Fig 5). Sublethal doses of compounds were chosen to preserve the overall parasite viability and pairing. For each compound, 3-5 couples in two independent experiments were analyzed and images of ootype, ovary, vitellarium, testis, gut, and parenchyma were recorded. The overall morphology of the ovary in worm pairs treated with NF2836 and NF2839 compounds seemed not to be affected thus retaining the organization with small immature oocytes (io) in the anterior part and large mature oocytes (mo) in the posterior part. However, a decrease of cellularity in the ovary mature compartment was observed in 5/11 samples treated with NF2836 at 50 uM; cell degeneration especially in the mature compartment was detected in the oocytes of all NF2839-treated worms, with an increase in the presence of black spots in the worms treated with 10 µM (Fig 5). The vitellarium retains a preserved structure in all analyzed samples. Regarding the ootype, worm pairs treated with both compounds at the higher concentration showed either an empty ootype or no properly developed eggs and/or egg fragments, while at the lower concentrations, 5/7 or 4/7 ootype contained damaged eggs respectively in NF2836- or NF2839-treated worms (Fig 5). In addition, treatment with NF2836 and NF2839 compounds at the higher concentration led in all worms, in particular males, to gut dilation with a dramatic decrease in the number of surface amplification and lumen epithelial thinning. Accumulation of carmine-red positive particle aggregates was also present in the lumen of the NF2839-treated male worms (Fig 5).

# Effects of the quinolone-based compounds on adult worm pairs histone and tubulin acetylation

To investigate the HDAC inhibitory activity in adult parasites, cytosolic and nuclear-enriched fractions of worm pairs treated for 48 h and 72 h with partially lethal and sub-lethal doses of NF2836, NF2838, and NF2839 compounds were analyzed by western blotting (Fig 6). The samples treated with the three compounds showed increased acetylation of both histone proteins and tubulin when compared with vehicle (DMSO). TSA, a known pan- class I and class II HDAC -inhibitor, and NF2503, a HDAC inhibitor previously shown to impact histone acetylation in adult worm pairs [18] were used as positive controls. NF2839 and NF2838 treatment more strongly impact histone acetylation, while NF2836, which only slightly impacts histone acetylation, induces the highest increase in tubulin acetylation (Fig 6).

#### Activity of hit compounds on recombinant SmHDAC8

All three quinolone-based compounds inhibited the activity of the recombinant form of SmHDAC8 at the low micromolar range (Table 1). Therefore, these HDAC inhibitors have a deacetylase inhibitory activity in parasites and are also targeting SmHDAC8. The increase in tubulin acetylation in cell lysates of HDAC inhibitors-treated worm pairs (Fig 6) and their nanomolar inhibitory activity on hHDAC6 (Table 1) point to the possibility that others Class I



Fig 5. Confocal microscopy of carmine-red stained adult worm pairs treated with quinolone-based HDAC inhibitors. The images are representative of 3–4 worm pairs treated with DMSO, NF2836 (20 and 50  $\mu$ M) or NF2839 (5 and 10  $\mu$ M) at 72 h in two independent experiments. The ovary, ootype and gut are shown. Immature oocytes (io), mature oocytes (mo), ootype (ot) and gastrodermis (ga) are labeled. In the Figure, red filled triangles indicate degenerated mo; empty red triangles indicate egg fragments in the ootypes; asterisks indicate deformed egg in the ootype and carmine red aggregates in the gut lumen, filled red arrows indicate the ga in the gut. Scale bars: 50  $\mu$ m.

https://doi.org/10.1371/journal.pntd.0011992.g005



**Fig 6. Effects of selected compounds on histones and tubulin acetylation.** Representative immunoblots of the histone-enriched and cytosolic protein fractions extracted from *S. mansoni* adult worm pairs incubated with anti-acetylated lysine antibody (acetylated-K) or anti-acetylated tubulin. Worm pairs were treated for 48 or 72 h at the concentration indicated for each compound. DMSO (vehicle) and of the HDAC pan- inhibitor, TSA (1 μM), were used respectively as negative and positive controls.

https://doi.org/10.1371/journal.pntd.0011992.g006

or II HDACs could be targeted. Interestingly non-selective benzanilide HDAC inhibitors with very little, to no-observed potency against both human and *S. mansoni* HDAC8 orthologues and schistosomicidal activity have been recently described [22]. Overall, we cannot exclude that the NF2839 and the other quinolone-based compounds are impacting parasite viability and/or eggs viability and maturation also through engagement of other HDACs. Moreover, off-targets effects cannot be excluded.

Table 1. Activity of quinoine-based compounds on similarce activity.					
HDAC Inhibitor	hHDAC1* (μM)	hHDAC6* (nM)	hHDAC8* (μM)	SmHDAC8 (µM)	
NF2836	$3.0 \pm 0.2$	33.0 ± 2.6	$1.6 \pm 0.3$	$2.3 \pm 0.8$	
NF2838	$0.3 \pm 0.1$	$11.5 \pm 1.5$	$0.5 \pm 0.2$	$0.4 \pm 0.03$	
NF2839	$0.3 \pm 0.1$	$6.9 \pm 0.9$	$0.6 \pm 0.2$	$0.1 \pm 0.3$	

Table 1. Activity of quinolone-based compounds on SmHDAC8 activity.

The  $IC_{50}$  of the indicated inhibitors on recombinant SmHDAC8 enzymatic activity are shown as an average of three independent experiments; uncertainties are given as standard deviations. The Fluor-de-Lys substrate was used at 50  $\mu$ M, and the inhibitor concentrations ranged from 25  $\mu$ M to 0.04  $\mu$ M. The  $IC_{50}$  on recombinant hHDAC1, hHDAC6 and hHDAC8 enzymatic activity previously reported [25] are also listed.

https://doi.org/10.1371/journal.pntd.0011992.t001

#### In silico prediction of selected drug-like properties and cytotoxicity evaluation of the hit compounds

The ADME properties and the heart toxicity potential of the compounds by means of QikProp software (Version 4.3, Schrodinger, LCC, New York, NY 2015) have been previously reported [25] and are summarized in <u>S1 Table</u>. The predicted values can be considered acceptable and were within the recommended range.

The cytotoxicity of the hit compounds was assessed in two independent experiments on murine L929 and human BJ fibroblasts at 24 h by MTT assays. Vehicle (DMSO) and Gambo-gic acid (GA) were used as negative and positive control respectively. The compounds show a safe profile on both cell lines (S2 Table).

### Conclusions

We discovered a schistosomicidal activity of a quinolone-based HDAC inhibitor, NF2839, on both larval and adult worm pairs also impacting egg viability and maturation *in vitro*. Importantly, this compound increases histone and tubulin acetylation in *S. mansoni* parasites, representing a candidate worthy of further consideration as starting point for the development of new generation anti-*Schistosoma* chemotherapeutics.

### **Supporting information**

S1 Data. Excel spreadsheet containing, in separate sheets, the underlying numerical data and statistical analysis for Figs <u>3A</u>, <u>3B</u>, and <u>4</u>. (XLSX)

S1 Table. Predicted drug-like features for NF2836, NF2838 and NF2839 compounds. <sup>a</sup>SASA predicted the total solvent accessible surface (range or recommended value for 95% of known drugs 300–1000); <sup>b</sup>QPlogP predicted octanol/water partition coefficient (range or recommended value for 95% of known drugs -2–6.5); <sup>c</sup>QPlogS predicted aqueous solubility in mol/dm<sup>3</sup> (range or recommended value for 95% of known drugs -6.5–0.5); <sup>d</sup>QPPCaco predicted apparent Caco-2 cell permeability in nm/sec (range or recommended value for 95% of known drugs >500 great); <sup>e</sup>QPPMDCK predicted apparent MDCK cell permeability in nm/ sec (range or recommended value for 95% of known drugs >500 great); <sup>f</sup>QPlogHERG predicted IC<sub>50</sub> values for blockage of HERG K<sup>+</sup> channels (range or recommended value for 95% of known drugs below -5); <sup>g</sup>%HOA predicted human oral absorption on 0 to 100% scale (range or recommended value for 95% of known drugs >80% high). Range or recommended values are reported in QikProp user manual. \*Cmpd name reported in Relitti et al [25]. (DOCX)

S2 Table. Cytotoxicity for NF2836, NF2838 and NF2839 compounds on BJ and L929 fibroblast cell lines. The  $IC_{50}(\mu M)$  shown was calculated by dose-response curves of the compounds on murine L929 and human BJ fibroblast cells. The range of concentrations were for NF2836, NF2838, NF2839: 100–0.78  $\mu$ M; and for GA 10–0.078  $\mu$ M. The data are representative of 2 independent experiments. Gambogic acid (GA) and vehicle (DMSO) were used as positive and negative controls respectively. (DOCX)

### Acknowledgments

We wish to thank Dr. Marialaura Giannaccari, IBBC-CNR for her support on SmHDAC8 recombinant protein purification, Stefania Colantoni for mouse husbandry and Pierluigi Palozzo for dishwashing and lab technical support.

#### **Author Contributions**

Conceptualization: Giovina Ruberti.

- **Data curation:** Roberto Gimmelli, Giuliana Papoff, Fulvio Saccoccia, Cristiana Lalli, Sandra Gemma, Giuseppe Campiani, Giovina Ruberti.
- **Formal analysis:** Roberto Gimmelli, Giuliana Papoff, Fulvio Saccoccia, Cristiana Lalli, Sandra Gemma, Giuseppe Campiani, Giovina Ruberti.

Funding acquisition: Giuseppe Campiani, Giovina Ruberti.

Investigation: Roberto Gimmelli, Giuliana Papoff, Fulvio Saccoccia, Cristiana Lalli, Sandra Gemma.

Methodology: Roberto Gimmelli, Giuliana Papoff, Fulvio Saccoccia, Cristiana Lalli.

Supervision: Giovina Ruberti.

Validation: Roberto Gimmelli, Giuliana Papoff, Fulvio Saccoccia.

Visualization: Roberto Gimmelli, Giuliana Papoff, Fulvio Saccoccia, Sandra Gemma.

Writing - original draft: Giovina Ruberti.

Writing – review & editing: Roberto Gimmelli, Giuliana Papoff, Fulvio Saccoccia, Sandra Gemma, Giuseppe Campiani, Giovina Ruberti.

#### References

- 1. McManus DP, Dunne DW, Sacko M, Utzinger J, Vennervald BJ, Zhou XN. Schistosomiasis. Nat Rev Dis Primers 2018; 4: 13–31. https://doi.org/10.1038/s41572-018-0013-8 PMID: 30093684
- Molehin AJ, McManus DP, You H. Vaccines for human schistosomiasis: recent progress, new developments, and future prospects. Int J Mol Sci. 2022; 23: 2255–2267. https://doi.org/10.3390/ijms23042255 PMID: 35216369
- Cioli D, Pica-Mattoccia L, Basso A, Guidi A. Schistosomiasis control: praziquantel forever? Mol Biochem Parasitol. 2014; 195: 232–239. https://doi.org/10.1016/j.molbiopara.2014.06.002 PMID: 24955523
- Borlase A, Rudge JW, Léger E, Diouf ND, Fall CB, Diop SD, et al. Spillover, hybridization, and persistence in schistosome transmission dynamics at the human-animal interface. Proc Natl Acad Sci USA 2011;12: 118(41):e2110711118: https://doi.org/10.1073/pnas.2110711118 PMID: 34615712
- Léger E, Webster JP. Hybridizations within the genus Schistosoma: implications for evolution, epidemiology and control. *Parasitology* 2017; 144: 65–80. https://doi.org/10.1017/S0031182016001190 PMID: 27572906
- Teukeng FFD, Blin M, Bech N, Gomez MR, Zein-Eddine R, Simo AMK, et al. Hybridization increases genetic diversity in *Schistosoma haematobium* populations infecting humans in Cameroon. Infect Dis Poverty 2022; 26: 11(1), 37. https://doi.org/10.1186/s40249-022-00958-0 PMID: 35346375
- Rothe C, Zimmer T, Schunk M, Wallrauch C, Helfrich K, Gültekin F, et al. Developing Endemicity of Schistosomiasis, Corsica, France. Emerg Infect Dis. 2021; 27(1): 319–321. <u>https://doi.org/10.3201/eid2701.204391</u> PMID: 33264582
- Fioravanti R, Mautone N, Rovere A, Rotili D, Mai A. Targeting histone acetylation/deacetylation in parasites: an update (2017–2020). Curr Opin Chem Biol. 2020; 57: 65–74. https://doi.org/10.1016/j.cbpa. 2020.05.008 PMID: 32615359
- Park SY, Kim, JS A short guide to histone deacetylases including recent progress on class II enzymes. Exp Mol Med 2020; 52: 204–212. https://doi.org/10.1038/s12276-020-0382-4
- Oger F, Dubois F, Caby S, Noel C, Cornette J, Bertin B, et al. The class I histone deacetylases of the platyhelminth parasite *Schistosoma mansoni*. Biochem Biophys Res Commun. 2008; 377: 1079–1084. https://doi.org/10.1016/j.bbrc.2008.10.090 PMID: 18977200
- Dubois F, Caby S, Oger F, Cosseau C, Capron M, Grunau C, et al. (2009) Histone deacetylase inhibitors induce apoptosis, histone hyperacetylation and up-regulation of gene transcription in *Schistosoma mansoni*. Mol Biochem Parasitol. 2009; 168: 7–15. https://doi.org/10.1016/j.molbiopara.2009.06.001
  PMID: 19538992

- Scholte LL, Mourão MM, Pais F, Melesina J, Robaa D, Volpini AC, et al. Evolutionary relationships among protein lysine deacetylases of parasites causing neglected diseases. Infect Genet Evol. 2017; 53: 175–188. https://doi.org/10.1016/j.meegid.2017.05.011 PMID: 28506839
- Marek M, Kannan S, Hauser AT, Moraes Mourao M, Caby S, Cura V, et al. Structural basis for the inhibition of histone deacetylase 8 (HDAC8), a key epigenetic player in the blood fluke *Schistosoma mansoni*. PLoS Pathog. 2013; 9: e1003645. https://doi.org/10.1371/journal.ppat.1003645 PMID: 24086136
- Chakrabarti A, Oehme I, Witt O, Oliveira G, Sippl W, Romier C, et al. HDAC8: a multifaceted target for therapeutic interventions. Trends Pharmacol Sci. 2015; 36: 481–492. https://doi.org/10.1016/j.tips. 2015.04.013 PMID: 26013035
- Guidi A, Saccoccia F, Gennari N, Gimmelli R, Nizi E, Lalli C, et al. Identification of novel multi-stage histone deacetylase (HDAC) inhibitors that impair *Schistosoma mansoni* viability and egg production. Parasites Vectors 2018; 11: 668. https://doi.org/10.1186/s13071-018-3268-8 PMID: 30587243
- Heimburg T, Chakrabarti A, Lancelot J, Marek M, Melesina J, Hauser A-T, et al. Structure-based design and synthesis of novel inhibitors targeting HDAC8 from *Schistosoma mansoni* for the treatment of schistosomiasis. J Med Chem. 2016; 59: 2423–2435. https://doi.org/10.1021/acs.jmedchem.5b01478 PMID: 26937828
- Marek M, Shaik TB, Heimburg T, Chakrabarti A, Lancelot J, Ramos-Morales E, et al. Characterization of Histone Deacetylase 8 (HDAC8) selective inhibition reveals specific active site structural and functional determinants. J Med Chem. 2018; 61: 10000–10016. https://doi.org/10.1021/acs.jmedchem. 8b01087 PMID: 30347148
- Saccoccia F, Brindisi M, Gimmelli R, Relitti N, Guidi A, Saraswati AP, et al. Screening and phenotypical characterization of *Schistosoma mansoni* Histone Deacetylase 8 (SmHDAC8) inhibitors as multistage antischistosomal agents. ACS Infect Dis. 2020; 6(1): 100–113. <u>https://doi.org/10.1021/acsinfecdis.</u> 9b00224 PMID: 31661956
- 19. Ghazy E, Abdelsalam M, Robaa D, Pierce RJ, Sippl W. Inhibitors for the treatment of Schistosomiasis. Pharmaceuticals 2022; 15: 80. https://doi.org/10.3390/ph15010080 PMID: 35056137
- Hassan MM, Sedighi A, Olaoye OO, Häberli C, Merz A, Ramos-Morales E, et al. Phenotypic screening of Histone Deacetylase (HDAC) inhibitors against *Schistosoma mansoni*. Chem Med Chem. 2022; 17 (18): e202100622. https://doi.org/10.1002/cmdc.202100622 PMID: 35983937
- Saccoccia F, Pozzetti L, Gimmelli R, Butini S, Guidi A, Papoff G, et al. Crystal structures of *Schistosoma mansoni* histone deacetylase 8 reveal a novel binding site for allosteric inhibitors. J Biol Chem. 2022; 298(10):102375. https://doi.org/10.1016/j.jbc.2022.102375 PMID: 35970392
- Noce B, Di Bello E, Zwergel C, Fioravanti R, Valente S, Rotili D, et al. (2023) Chemically diverse S. mansoni HDAC8 inhibitors reduce viability in worm larval and adult stages. Chem Med Chem. 2023; 18(3): e202200510. https://doi.org/10.1002/cmdc.202200510 PMID: 36250286
- Campiani G, Cavella C, Osko JD, Brindisi M, Relitti N, Brogi S et al. Harnessing the role of HDAC6 in idiopathic pulmonary fibrosis: design, synthesis, structural analysis, and biological evaluation of potent inhibitors. J Med Chem. 2021; 64(14): 9960–9988. https://doi.org/10.1021/acs.jmedchem.1c00184 PMID: 34251197
- Saraswati AP, Relitti N, Brindisi M, Osko JD, Chemi G, Federico S, et al. Spiroindoline-capped selective HDAC6 Inhibitors: design, synthesis, structural analysis, and biological evaluation. ACS Med Chem Lett. 2020; 11(11): 2268–2276. https://doi.org/10.1021/acsmedchemlett.0c00395 PMID: 33214839
- Relitti N, Saraswati AP, Chemi G, Brindisi M, Brogi S, Herp D, et al. Novel quinolone-based potent and selective HDAC6 inhibitors: synthesis, molecular modeling studies and biological investigation. Eur J Med Chem. 2021; 212: 112998. https://doi.org/10.1016/j.ejmech.2020.112998 PMID: 33199154
- 26. Federico S, Khan T, Fontana A, Brogi S, Benedetti R, Sarno F, et al. Azetidin-2-one-based small molecules as dual hHDAC6/HDAC8 inhibitors: investigation of their mechanism of action and impact of dual inhibition profile on cell viability. Eur J Med Chem. 2022; 238: 114409. <u>https://doi.org/10.1016/j.ejmech.</u> 2022.114409 PMID: 35551034
- Mosmann T. Rapid colorimetric assay for cellular growth and survival: Application to proliferation and cytotoxicity assays. J. Immunol. Meth. 1983; 65: 55–63. https://doi.org/10.1016/0022-1759(83)90303-4 PMID: 6606682
- Presutti D, Santini S, Cardinali B, Papoff G, Lalli C, Samperna S, et al, MET Gene Amplification and MET Receptor Activation Are Not Sufficient to Predict Efficacy of Combined MET and EGFR Inhibitors in EGFR TKI-Resistant NSCLC Cells. PloS One 2015; 10:e0143333. <u>https://doi.org/10.1371/journal.pone.0143333</u> PMID: 26580964
- 29. Lalli C, Guidi A, Gennari N, Altamura S, Bresciani A, Ruberti G. Development and validation of a luminescence-based, medium-throughput assay for drug screening in *Schistosoma mansoni*. PLoS Negl Trop Dis. 2015; 9: e0003484. https://doi.org/10.1371/journal.pntd.0003484 PMID: 25635836

- **30.** Guidi A, Gimmelli R, Bresciani A, Ruberti G. Luminescence-based, low- and medium-throughput assays for drug screening in *Schistosoma mansoni* larval stage. Met Mol Biol. 2020; 2151: 219–227.
- Guidi A, Lalli C, Gimmelli R, Nizi E, Andreini M, Gennari N, et al. Discovery by organism based highthroughput screening of new multi-stage compounds affecting *Schistosoma mansoni* viability, egg formation and production. PLoS Negl Trop Dis. 2017; 11: e0005994. https://doi.org/10.1371/journal.pntd. 0005994 PMID: 28985236
- Jurberg AD, Gonçalves T, Costa TA, de Mattos AC, Pascarelli BM, de Manso PP, et al. The embryonic development of *Schistosoma mansoni* eggs: proposal for a new staging system. Dev Genes Evol. 2009; 219(5): 219–234. https://doi.org/10.1007/s00427-009-0285-9 PMID: 19415326
- **33.** Michaels RM, Prata A. Evolution and characteristics of *Schistosoma mansoni* eggs laid *in vitro*. J Parasitol 1968; 54: 921–930. https://doi.org/10.2307/3277120